Author's response to reviews

Title: Classifying perinatal mortality using verbal autopsy: Is there a role for non-physicians?

Authors:

- Cyril Engmann (cengmann@med.unc.edu)
- John Ditekemena (josol70@yahoo.fr)
- Imtiaz Jehan (imtiaz.jehan@aku.edu)
- Ana Garces (anagarces@imsalud.org)
- Mutinta Phiri (drtinta@yahoo.com)
- Vanessa Thorsten (vthorsten@rti.org)
- Manolo Mazariegos (mmazariegos@incap.int)
- Elwyn Chomba (echomba@zamnet.zm)
- Omrana Pasha (omrana.pasha@aku.edu)
- Antoinette Teshefu (antotshe@yahoo.com)
- Elizabeth McClure (mcclure@rti.org)
- Dennis Wallace (dwallace@rti.org)
- Robert Goldenberg (rgoldenb@drexelmed.edu)
- Waldemar Carlo (Wcarlo@Peds.UAB.EDU)
- Linda Wright (wrightl@exchange.nih.gov)
- Carl Bose (cbose@med.unc.edu)

Version: 5  Date: 28 June 2011

Author's response to reviews:

Dear Editors,

Thank you for extending the deadline for submission of this manuscript to June 27 so we could attend to the comments raised by the reviewers and Editorial team. Please find enclosed a revised version of the manuscript which incorporates the confidence intervals for the relative difference calculations and the inclusion of the kappa statistic. We have highlighted in yellow the text that accompanies these two additions.

Regards,

Cyril Engmann MD FAAP
Attending Neonatologist
Associate Professor
Departments of Pediatrics and Maternal Child Health
University of North Carolina Schools of Medicine and Public Health
CB# 7596, 4th Floor UNC Hospitals
The University of North Carolina at Chapel Hill Chapel Hill, NC 27599-7596
Phone 919-9665063
Pager 919-2163241
Fax 919-9663034
Email cengmann@med.unc.edu